Author's response to reviews

Title: An ongoing case-control study to evaluate the NHS Bowel Cancer Screening Programme

Authors:

Nathalie J Massat (n.massat@qmul.ac.uk)
Peter D Sasieni (p.sasieni@qmul.ac.uk)
Dharmishta Parmar (d.parmar@qmul.ac.uk)
Stephen W Duffy (s.w.duffy@qmul.ac.uk)

Version: 2 Date: 1 December 2014

Author's response to reviews: see over
01 December 2014

Dear Dr Solera,

Massat et al, An ongoing case-control study to evaluate the NHS Bowel Cancer Screening Programme

At the demand of the editorial board, the following changes have been made to the original submission dated 21.11.14:

1) Additional files have been removed;

2) References have been reformatted within the text to "[X]".

The protocol has received full ethics approval and relevant documentation is attached to the submission.

I look forward to hearing from you.

Yours ever,

Stephen W. Duffy, Professor of Cancer Screening